Author's response to reviews

Title: Building effective service linkages in primary mental health care: a narrative review part 2

Authors:

Jeffrey D Fuller (jeffrey.fuller@flinders.edu.au)
David Perkins (dperkins@gwahs.health.nsw.gov.au)
Sharon Parker (smparker@bigpond.net.au)
Louise Holdsworth (louise.holdsworth@scu.edu.au)
Brian Kelly (brian.kelly@newcastle.edu.au)
Russell Roberts (rroberts@gwahs.health.nsw.gov.au)
Lee Martinez (lee.martinez@health.sa.gov.au)
Lyn Fragar (lyn.fragar@hnehealth.nsw.gov.au)

Version: 5 Date: 1 February 2011

Author's response to reviews: see over
Part 1 paper

Reviewer Peter Bower

Minor essential

The abstract has been edited to make clear the distinction between the concept of partnership formation, that came from the literature, and a related but different set of proposed 5 strategies that were made by the expert reference group (see highlighted sections in the abstract).

Reviewer Rachel Jenkins

As per comments re paper 1